Contents lists available at ScienceDirect

# **Renewable Energy**



# The impact of climate policy uncertainty on renewable and non-renewable energy demand in the United States

Yunfeng Shang<sup>a</sup>, Ding Han<sup>b</sup>, Giray Gozgor<sup>c,d,e,\*</sup>, Mantu Kumar Mahalik<sup>f</sup>, Bimal Kishore Sahoo<sup>f</sup>

<sup>a</sup> School of Hospitality Administration, Zhejiang Yuexiu University, Shaoxing, China

<sup>b</sup> International Business Strategy Institute, University of International Business and Economics (UIBE), Beijing, China

<sup>c</sup> School of Management, University of Bradford, Bradford, United Kingdom

<sup>d</sup> Faculty of Political Sciences, Istanbul Medeniyet University, Istanbul, Turkey

<sup>e</sup> Adnan Kassar School of Business, Lebanese American University, Beirut, Lebanon

<sup>f</sup> Department of Humanities and Social Sciences, Indian Institute of Technology Kharagpur, Midnapur, 721302, West Bengal, India

#### ARTICLE INFO

Keywords: Climate policy uncertainty Renewable energy demand Non-renewable energy demand Economic growth Crude oil price

#### ABSTRACT

This paper evaluates the impact of climate policy uncertainty on renewable and non-renewable energy consumption in the United States over the quarterly data from 2000Q1 to 2021Q3. Economic growth and crude oil prices are added to the energy consumption functions as control variables. The paper considers several approaches to model both renewable and non-renewable energy demand. It is found that crude oil prices promote non-renewable energy demand and climate policy uncertainty reduces it. Surprisingly, the impact of economic growth on non-renewable energy consumption is positive but insignificant. It is also observed that economic growth promotes renewable energy demand, and crude oil prices reduce it. Furthermore, climate policy uncertainty positively affects renewable energy demand in the long run. Some policy implications are provided for reducing non-renewable energy consumption and promoting renewable energy use in the United States through climate policy implementation.

## 1. Introduction

Climate change is one of the leading problems in the 21st century. Scholars have demonstrated that risks and uncertainties related to climate change can affect various dimensions of the global economic system [1,2]. Climate change is attributed to global warming, environmental pollution, and caused by greenhouse gas emissions. Also, CO2 emissions are driven by fossil fuels and non-renewable energy sources [3,4]. Therefore, the transition from fossil fuels to green renewable energy is an important policy tool to transfer economies to the low carbon economy. It is suggested that higher renewable energy consumption can slow down the negative consequences of climate change on the economic system [5,6]. Similarly [7], argued that promoting renewable energy usage in consumption and production activities can also reduce the adverse impact of climate change and global warming on living human beings and natural habitats in the long run. Therefore, the determinants of the non-renewable and renewable energy demand are crucial for designing climate change policies in developing and

developed economies.

Since the early 1990s, the transformation from non-renewable to renewable energy has increased in advanced countries and some developing economies. This transformation to renewable and other alternative energy sources has four aspects. The first reason is the progress in technology, and the costs of new investments in energy sources have been significantly reduced [8-10]. The second issue is governments' renewable energy supporting policies (e.g., defining portfolio standards, providing cheaper credits and tax benefits on renewable energy investments) [11–13]. The third aspect is the climate change crisis. Negative outcomes of climate change are related to greenhouse gas emissions. Fossil fuels (non-renewable energy sources) increase greenhouse gas emissions. In other words, renewable energy sources can decrease greenhouse gas emissions, favouring the slowdown of climate change [14–16]. Finally, the price volatility in crude oil due to the geopolitical risks and market uncertainty has increased the interest in alternative energy sources [17,18].

Given this backdrop, this paper analyses the determinants of the non-

https://doi.org/10.1016/j.renene.2022.07.159

Received 13 April 2022; Received in revised form 9 June 2022; Accepted 25 July 2022 Available online 7 August 2022

0960-1481/© 2022 The Authors. Published by Elsevier Ltd. This is an open access article under the CC BY license (http://creativecommons.org/licenses/by/4.0/).





<sup>\*</sup> Corresponding author. School of Management, University of Bradford, Bradford, United Kingdom.

*E-mail addresses*: 20141075@zyufl.edu.cn (Y. Shang), 202101930165@uibe.edu.cn (D. Han), g.gozgor@bradford.ac.uk (G. Gozgor), mkm@hss.iitkgp.ac.in (M.K. Mahalik), bks@hss.iitkgp.ac.in (B.K. Sahoo).

renewable and renewable energy demand. The paper focuses on the quarterly 21st century data of the largest  $CO_2$  emitter economy in the World, i.e., the United States, from 2000Q1 to 2021Q3. Note that the United States is also the largest primary energy consumer in the World, with a share of 15.8% in 2020, and the share of  $CO_2$  emission is 13.8% in 2020 [19]. The country's electricity generation from renewable energy is increasing yearly, and this transformation in the energy mix decreases carbon dioxide emissions [61]. Despite this effort, Fig. 1 shows that the carbon emissions are increasing in the United States compared to other emerging economies of the World from 2012 to 2018.

The United States economy is chosen for our analysis because it is one of the advanced economies where fossil fuel consumption per capita is 60,167 kWh in 2020, keeping the United States as the number one country, followed by China is 23,674 kWh (4th), and India is 5, 7888 kWh (9th).<sup>1</sup> It shows that the economy like the United States is lagging behind other growing countries in increasing renewable energy investment and reducing fossil fuel consumption and carbon emissions. Moreover, renewable energy investment as % of gross domestic product (GDP) in 2015 was 0.2% for the United States, which is much lower than other emerging countries, i.e. South Africa (1.4%), China (0.9%), India (0.5%), and Brazil (0.4%). Therefore, when the World is facing the threat of climate change and global warming, our study is timing one and motivates us to explore the determinants of non-renewable and renewable energy demand for the United States so that better climate change mitigating policy could be designed for sustainable environment in long-run.

This paper uses a new potential energy demand driver, i.e., the index of climate policy uncertainty introduced by Ref. [20]. The Climate Policy Uncertainty index is similar to the Economic Policy Uncertainty index. Precisely, it measures the changes in government policies on environmental issues (e.g., changes in subsidiaries and tax regulations). The policy changes can create uncertainties that can delay consumption and investment decisions by consumers and firms. For example, uncertainty about subsidiaries and tax regulations can create extra costs for the firms. Therefore, climate policy uncertainty can affect firms by delaying or postponing investments and reducing economic output. The possible changes in energy consumption and investments can also affect economic performance [21].

As we discuss in the next section, few papers in the empirical literature consider the determinants of renewable energy demand in the United States [22]; Schumacher and [23]; Yang, 2018) but ignores the role of Climate Policy Uncertainty index of [20] on non-renewable and renewable energy demand. However, to the best of our knowledge, there is no paper in the literature to consider the Climate Policy Uncertainty index of [20] in the renewable and non-renewable energy demand functions for the United States. This paper contributes to the empirical literature by using the climate policy uncertainty index as a new measure of environmental policy or regulation affecting non-renewable and renewable energy demand. Both economic growth and oil price are included in renewable and non-renewable energy demand functions as control variables, following the existing studies (Sadosrky et al., 2009a, 2009b; [7,24-26]. The paper uses various time-series techniques to focus on the case of the United States with the quarterly data from 2000Q1 to 2021Q3. The paper shows that climate policy uncertainty reduces non-renewable energy demand. Furthermore, climate policy uncertainty positively affects renewable energy demand in the long run. Therefore, climate policy uncertainty is crucial in changing the energy demand in the United States economy in the long run. This evidence is reflected in Fig. 2 & 3, where one can see the long-run relation between the climate policy uncertainty and the pattern of non-renewable and non-renewable energy demand in the United States. Interestingly, the relationship between climate policy uncertainty and renewable energy consumption appears close to that between non-renewable energy

consumption and climate policy uncertainty in the United States. This issue is another motivation for us to understand whether climate policy uncertainty equally promotes the renewable and non-renewable energy demand in the United States.

The rest of the paper is organized as follows. Section 2 reviews the previous papers in the empirical literature for the determinants of energy demand and the impact of climate change uncertainty on energy variables. Section 3 explains the details of the data and the methodology. Section 4 discusses the empirical results and their implications. Section 5 highlights policy implications. Section 6 concludes with evaluates limitations and future research.

#### 2. Literature review

#### 2.1. Determinants of energy demand

There are various papers to analyse the determinants of energy demand. Among the earliest studies [17], examines the determinants of renewable energy in the G7 countries and finds that per capita income and CO2 emissions increase renewable energy, but oil price harms it. In a further study [18], illustrates that the per capita income positively relates to renewable energy in 18 emerging economies. However, [24]; using the panel technique for 24 European countries, found that income level and carbon dioxide emissions are the important drivers of renewable energy demand. Their findings indicate that European countries can reduce their dependence on fossil fuels (i.e. coal, oil, and natural gas) if they can invest more in renewable energy deployment with the increased income level support.

[25] observe the impacts of income, carbon dioxide emissions and crude oil price on renewable energy consumption in six major emerging economies (i.e. Brazil, China, India, Indonesia, Philippines, and Turkey). They indicate that the crude oil price is the main determinant of the renewable energy demand in China and Indonesia, where oil price reduces the adoption of clean energy. They also find the positive impact of per capita income on renewable energy demand in all six emerging economies, while pollutant emissions increase the renewable energy in Brazil, China, India, and Indonesia). Their findings provide valuable insights for emerging economies to reduce carbon intensity by improving the share of renewable energy in the primary energy mix [26]. investigate the drivers of renewable energy demand in 64 countries from 1990 to 2011. They find that income level and carbon emissions positively and significantly impact renewable energy demand, whereas oil prices harm it.

Similarly [27], study the determinants of renewable energy demand in 38 countries. Carbon emissions and income levels are significant drivers of renewable energy growth. [22]; using 1990–2008 state-level panel data for the United States' electricity market, studied the reasons for the development of renewable energy and found that promoting renewable energy projects as a potential job creator is one of the main drivers of renewable energy projects. Lin and Moubarak [60] show that economic output and financial development positively affect renewable electricity consumption, while trade openness, FDI inflows and traditional energy lobby activities reduce it in China.

Meanwhile [28], indicate that domestic and foreign capitals promote clean energy use in the EU, the G20, and the OECD countries from 1993 to 2012. [29]; using China's regional data, finds that economic development enhances renewable energy production [30]. shows that economic freedom positively affects renewable energy demand in 28 member states. Schumacher and Yang [31], using the state-level data in the United States, find that financial incentives and regulatory measures are weak indicators of wind energy growth [7]. indicate that economic globalization increases the renewable energy demand in the panel dataset of 30 OECD countries from 1970 to 2015. They further indicate the positive effects of per capita income, oil price and per capita carbon emissions on renewable energy for a sample of 30 OECD economies. Their findings provide insightful insights for advanced economies where

<sup>&</sup>lt;sup>1</sup> https://ourworldindata.org/grapher/fossil-fuels-per-capita.



Fig. 1. The trend of CO<sub>2</sub> emissions per capita in the United States, India, China, and Brazil.



Fig. 2. The trend of non-renewable energy consumption and climate policy uncertainty in the United States.

they can improve the quality of the natural environment by increasing the share of renewable energy in the total primary energy mix with the help of income generation and increased oil price.

In addition [32], studied the determinants of renewable energy consumption using the panel data from 1995 to 2015 for a sample of 97 countries. They find that in countries where better democratic rights of the people are preserved, higher economic growth increases the use of renewable energy and vice-versa for less democratic countries. They also find that real oil prices induce renewable energy demand in less democratic countries and play no significant role in more democratic countries. Their findings show that democratic institutions are crucial in channelling economic growth to renewable energy [33]. examine the determinants of the energy demand in the panel dataset of 25 OECD economies from 1978 to 2016. The authors utilised various panel data estimation techniques and observed that real per capita income increases the energy demand. In addition, real energy prices and economic complexity decrease the energy demand. The authors conclude that technological progress measured by economic complexity decreases the necessity of energy consumption [34]. analyse the effects of age dependency and urbanisation on renewable and non-renewable energy consumption in Brazil, India, China, and South Africa from 1990 to 2019. The authors observe that age dependency ratios and urbanisation reduce the renewable and non-renewable energy demand. However,

economic growth promotes renewable and non-renewable energy demand. It is also found that globalisation (measured by FDI) has a mixed impact on inflows on energy demand in the related economies.

[23]; using the annual data from 1990 to 2020 for the United States, study the drivers of energy efficiency. They find that investment in renewable energy sources benefits energy efficiency, while industrial production, trade openness and financial inclusion improve it. [35]; using the panel data from 1993 to 2018 for a sample of 20 OECD countries, examine the role of energy prices and economic growth in renewable energy capacity expansion. They indicate economic growth and coal oil prices drive renewable energy capacity development. They also find the positive impact of natural gas prices on renewable energy capacity. Also, they found a positive impact of coal oil price on renewable energy capacity development than the price of natural gas. Their findings suggest having a quick transition to renewable energy through price effect, which is key to sustainable development [36]. studied the impact of higher oil prices, carbon emissions, and income on renewable energy consumption in resource-rich countries like Iran from 1980 to 2019. They indicate that oil price and carbon emissions have significant and negative impacts on renewable energy consumption, while income does not significantly impact it.

Few papers have considered uncertainty measures as a possible driver of the energy demand [37–39]. For instance Ref. [37], examine



Fig. 3. The trend of renewable energy consumption and climate policy uncertainty in the United States.

the impact of economic policy uncertainty (EPU) on renewable energy consumption in the USA using the monthly data from 1986 to 2019. They find that higher economic policy uncertainty lowers renewable energy consumption and vice-versa [38]. studies the asymmetric impact of policy-induced uncertainty on renewable energy consumption in G7 countries from 1997 to 2019. As control variables, income, institutions, and innovation are included in renewable energy consumption. The non-linear autoregressive lag model results indicate that both a negative and a positive shock in lagged policy uncertainty reduce renewable energy consumption in the long run. It shows that policy uncertainty is also a significant driver of renewable energy demand but not asymmetric in the short run. It is further found that income and investment in research & development drive renewable energy consumption in the long run. Their findings suggested promoting renewable energy use by reducing policy uncertainty and improving technological advances from a policy perspective.

Similarly [39], investigate the impact of the EPU on the per capita energy consumption in a panel dataset of 72 developing and developed economies from 1960 to 2016. The novel finding of the paper is that the EPU spurs energy consumption. Several econometric methods show that per income increases energy demand, but energy prices reduce it. These findings are robust enough to utilize different econometric techniques and exclude various countries from the dataset.

# 2.2. Effects of climate policy uncertainty measure on energy and environmental pollution indicators

The index of Climate Policy Uncertainty is a new indicator to measure changes in environmental policies, particularly in the United States. It may also measure the level of technology uncertainty related to environmental pollution. There are a few papers in the empirical literature for using the Climate Policy Uncertainty index of [20] on energy and environmental pollution indicators. For instance Ref. [40], investigate the impact of climate policy uncertainty of [20] on the return performances of brown and green energy stocks. It is observed that the index of climate policy uncertainty increases the returns of green stocks compared to brown stocks, especially during crisis periods.

[41] show the significant effects of the indices of the Economic Policy Uncertainty in China and Climate Policy Uncertainty of [20] on the price volatility of the Wind Carbon-Neutral Concept (CNCI) index. The Climate Policy Uncertainty has a higher forecast ability for the CNCI values during volatile market times [42]. investigates the impact of the index of the Climate Policy Uncertainty on Research and Development (R&D) investments of various firms in the United States from 2000 to 2019. There is a significant positive impact of the Climate Policy Uncertainty on the R&D investments of firms in general. However, the Climate Policy Uncertainty harms the R&D investments of heavy-emitter firms in the United States. [43]; using the monthly data from January 1988 to August 2017, study the impact of global energy market uncertainty and economic policy uncertainty on oil prices. They find that oil price variation is primarily contributed by the energy market uncertainty shock than the economic policy uncertainty shock following 12 months. It clearly shows that since oil price changes respond more to the energy market uncertainty, consumers and producers need to consider the shock coming from the energy market uncertainty before buying the oil from the open market. Otherwise, non-renewable energy consumers and producers will pay more to mitigate their consumption and production activities while buying the oil needed. Their findings also provide valuable insights into the policymakers' need to consider different types of uncertainty in the energy demand framework.

Overall, reviewing the existing studies indicates that there are various papers on the determinants of energy demand (Sadosrky et al., 2009a, 2009b; [7,24–26,35–39], but there are a few papers [40–42] in the empirical literature to consider the role of Climate Policy Uncertainty index of [20] on green stocks and innovation. This paper contributes to the empirical literature by using this new measure as a potential determinant of the non-renewable and renewable energy demand in the United States, with the quarterly data spanning from 2000Q1 to 2021Q3. The climate policy uncertainty reduces non-renewable energy demand. Moreover, the climate policy uncertainty positively affects renewable energy demand in the long run.

# 3. Data and methodology

# 3.1. Data

The paper focuses on the United States economy over the quarterly data from 2000Q1 to 2021Q3. The quarterly data period from 2000Q1 to 2021Q3 was chosen. Note that climate policy uncertainty data are only available for the United States. The quarterly data with a higher frequency enables us to effectively understand the relationships between the variables. Moreover, quarterly information increases the sample size and captures the significant policy changes in domestic and globalized countries related to mitigating climate change and achieving energy conservation [44].

The dependent variable is the source's primary energy consumption (quadrillion Btu). We divide primary energy consumption into nonrenewable energy (*Non-REC*) and renewable energy consumption (*REC*). The related data are downloaded from the US Energy Information Administration [62].

We use two control variables. The first captures the income effect on the energy demand: gross domestic product (GDP) per capita in the USD with the seasonally-adjusted annual rate (*LNGDPC*). Secondly, we control the price effect, measured by crude oil prices, based on the West Texas Intermediate (*WTI*)-Cushing, Oklahoma, USD per barrel. These data are obtained from the St. Louis Fed [45].

The main variable of interest is the United States Climate Policy Uncertainty (CPU) index introduced by Ref. [20]. Following the methodology of the economic policy uncertainty (EPU) index of Baker et al. (2016) [20], constructs the CPU index by searching words in eight leading United States, such as "uncertainty", "carbon dioxide", "climate risk", "greenhouse gas emissions", "CO<sub>2</sub> emissions" "global warming", "climate change", "green energy", "renewable energy" or "environmental", including the variants such as "uncertainties", "regulatory", "policies". These eight series are standardised to have a unit standard deviation and then averaged across newspapers by the period. Finally, the averaged series are normalised to have a mean value of 100 for the corresponding period under concern [20]. scales the number of relevant articles per month with each newspaper's total number of articles. The CPU index data are downloaded from the website https://www.policyu ncertainty.com/climate\_uncertainty.html.

Since we use the quarterly data for the United States, it is important to check the seasonality of the variables. Except for real GDP per capita, all remaining variables (i.e. NON-REC, REC, WTI, and CPU) possess seasonality in the data series. The seasonality present in the data series is adjusted with the help of additive and multiplicative decompositions. The seasonality adjusted series are considered for our empirical modelling. However, the values of other variables are small, have minimal variation and are highly predictable, which is not the case for real GDP per capita. Therefore, we use the natural logarithm to real GDP per capita to reduce skewness and bring homogeneity in data series values [44].

# 3.2. Conceptual framework

Existing studies identify various determinants of energy demand (Sadosrky et al., 2009a, 2009b; [7,24–26,35–39]. The results are inconclusive due to the missing important factors in energy demand modelling. Therefore, the key purpose of this analysis is to investigate the effects of climate policy uncertainty on renewable and non-renewable energy in the United States by controlling the economic growth and crude oil prices. Adding economic growth and crude oil prices in modelling renewable and non-renewable energy consumption helps us avoid omitted variable bias. To do so, we have employed the following equations for the United States economy within time series estimations:

$$NON - REC_t = f(LNGDPC_t, WTI_t, CPU_t)$$
(1)

$$REC_t = f(LNGDPC_t, WTI_t, CPU_t)$$
(2)

where  $NON - REC_t$  and  $REC_t$  stand for non-renewable energy consumption and renewable energy consumption at period t. LNGDPC<sub>t</sub> measures real GDP per capita used as a proxy for income level.  $WTI_t$ refers to the West Texas Intermediate used as a proxy for crude oil prices.  $CPU_t$  stands for climate policy uncertainty index. It is also essential for someone to understand the effects of income, crude oil price, and climate policy uncertainty on non-renewable and renewable energy consumption. Income can drive the demand for non-renewable and renewable energy. One can argue that the rising income of the people can enable them to buy renewable and non-renewable energy from the open energy market because it is required to carry out their consumption and production activities. Thus, we hypothesize that the higher the income, the higher the renewable and non-renewable energy demand. Crude oil price is also another determinant of non-renewable and renewable energy demand. Crude oil prices can increase the demand for non-renewable energy if renewable energy price becomes expensive for the people in the open energy market. In addition, crude oil prices can reduce renewable energy consumption if the government provides oil subsidies to cover the cost of oil producers and consumers. Finally, climate policy uncertainty can reduce the demand for non-renewable energy. As we know, producers often want profit maximization at the cost of the natural environment. Such a situation calls for stringent environmental regulation imposed on the producers if they use dirty energy in their business, which generates pollution and threatens the balance of the eco-system. Therefore, a strict climate change mitigating policy can reduce the non-renewable energy demand. Also, stringent climate change policy can increase the demand for renewable energy. This issue is because people want to escape the pollution tax imposition under the climate policy regulation. One can conclude that climate policy uncertainty is the crucial determinant of non-renewable and renewable energy demand.

Since we aim to understand renewable and non-renewable energy consumption determinants, we have created separate energy consumption Models (Model1 & Model2) for our empirical setting. To be used for empirical estimation, Equations (1) and (2) are converted to econometric format in the form of intercept, the coefficients and error terms as specified in Eq. (3) and Eq. (4). Equations (3) and (4) indicate that variables are in a semi-logarithm form where the real GDP per capita is only converted into the logarithm. The data converted to their natural logarithm form reduces the skewed distribution of the variables and enables us to get the best estimates of the renewable and non-renewable energy consumption modelling.

Model 1 : 
$$NON - REC_t = \alpha_1 + \alpha_2 LNGDPC_t + \alpha_3 WTI_t + \alpha_4 CPU_t + \varepsilon_{1t}$$
(3)

Model 2: 
$$REC_t = \alpha_1 + \alpha_2 LNGDPC_t + \alpha_3 WTI_t + \alpha_4 CPU_t + \varepsilon_{2t}$$
 (4)

In Model1,  $\varepsilon_{1t}$  is the error term with the satisfying property of the normal distribution and  $\alpha_1, \alpha_2, \alpha_3, \alpha_4$  are the intercept, coefficients of real economic growth, crude oil prices, and climate policy uncertainty. A similar interpretation also applies to Model2. However, *LN* is the natural log taken for real GDP per capita to reduce the skewness in data series values.

# 3.3. Econometric procedures

#### 3.3.1. Unit root tests

In time series analysis, preventing the problem of spurious regression is essential [46]. Hence, the unit root was examined first with the help of Augmented Dickey and Fuller (ADF) [47] and the Phillips and Perron (PP) [48] tests. We account for the criticism levied at both tests regarding their size vulnerability, poor strength and incapacity to consider breaks in the series. The used variables have been further applied to a structural break test proposed by Ref. [49] to capture the structural changes arising in the data series. The linear Autoregressive Distributed Lag (ARDL) model accommodates a single structural break occurring in the data series. The method can also avoid the endogeneity arising from the link between independent variables and error terms while estimating the carbon dioxide emissions models.

#### 3.3.2. ARDL model

This study uses the ARDL bound test [50,51] to investigate the presence of long-term cointegration among the variables. The significant benefits of the ARDL bound testing approach over other cointegration approaches, such as [52]; are that it does not really place any constraint on data of any type and also estimates the energy demand models in the presence of mixed order of integration, i.e. I (1) or I (0). The usage of this model can be handicapped if the series are found to be integrated at I(2). It is also an ideal model for small size samples and addresses the issue of

endogeneity associated with the linkage between independent variables and the error term.

The ARDL bounds testing approach's unrestricted error correction models (UECM) for the first model shown in above Eq. (3) and Eq. (4) can also be defined in Eq. (5):

$$\Delta NON - REC_{t} = \theta_{1} + \theta_{2} \text{LNGDPC}_{t-1} + \theta_{3} WTI_{t-1} + \theta_{4} CPU_{t-1}$$

$$+ \sum_{i=1}^{p} \alpha_{2i} \Delta NON - REC_{t-i} + \sum_{i=0}^{m} \alpha_{3i} \Delta \text{LNGDPC}_{t-i}$$

$$+ \sum_{i=0}^{n} \alpha_{4i} \Delta WTI_{t-i} + \sum_{i=0}^{k} \alpha_{5i} \Delta CPU_{t-i} + \varepsilon_{1t}$$
(5)

where  $\Delta$  represents lag operator;  $\theta_1$  is the constant;  $\varepsilon_{1t}$  symbolises the error term by satisfying the normal distribution's zero mean and constant variance. Moreover, the error term includes other non-renewable and renewable energy determinants assumed to be neutral. The first half of Eq. (5) deals with the long-run dynamics, whereas the second half with the summation sign also stands for the short-run dynamics between the variables. The ARDL bounds testing approach of the long-run cointegrating relationship is selected based on the F-test values based on the coefficient of the variables at their lagged levels. Hence, the null hypothesis of no cointegration relationship between the variables shown in Eq. (5) is estimated as  $H_0: \theta_2 = \theta_3 = \theta_4$  against the alternative hypothesis of the long-run relationship as  $H_1: \theta_2 \neq \theta_3 \neq \theta_4$ .

Suppose measured F-statistics surpass upper critical thresholds, i.e., lower and upper bounds critical values. In that case, the null hypothesis of no cointegration can be rejected. Therefore, the long-run relationship between the variables exists. The null hypothesis cannot be opposed if the F-statistics calculated are lower than the lower bound critical limits, enabling us to conclude that there is no long-term relationship between the variables. Suppose the calculated value of F-statistics lies between the lower and upper limits of the critical values. In that case, there can be no decisive conclusion on cointegration between the variables. If a long-term relationship is formed in Eq. (5), then the long-run relationship and related short-run dynamic error correction models (ECM) can be obtained using Eq. (6).

$$\Delta NON - REC_{t} = \theta_{1} + \sum_{i=1}^{p} \alpha_{2i} \Delta NON - REC_{t-i} + \sum_{i=0}^{m} \alpha_{3i} \Delta \text{LNGDPC}_{t-i} + \sum_{i=0}^{n} \alpha_{4i} \Delta WTI_{t-i} + \sum_{i=0}^{k} \alpha_{5i} \Delta CPU_{t-i} + \gamma_{1}ECM_{t} + \varepsilon_{1t}$$
(6)

where  $\gamma_1$  is the coefficient of system adjustment speed to long-term equilibrium between the variables. Finally, we use the Akaike Information Criterion (AIC) to select lag order in the ARDL model. Similarly, the second model is shown in above Eq. (4) for estimating both the long run and short-run relationships between the variables is also specified in Eq. (7) and Eq (8). The estimation process of the ARDL model for the following Eq. (7) and Eq. (8) is the same as the above. The long-run form is specified in Eq. (7) as follows:

$$\Delta REC_{t} = \theta_{1} + \theta_{2} \text{LNGDPC}_{t-1} + \theta_{3} WTI_{t-1} + \theta_{4} CPU_{t-1} + \sum_{i=1}^{p} \alpha_{2i} \Delta REC_{t-i}$$

$$+ \sum_{i=0}^{m} \alpha_{3i} \Delta \text{LNGDPC}_{t-i} + \sum_{i=0}^{n} \alpha_{4i} \Delta WTI_{t-i} + \sum_{i=0}^{k} \alpha_{5i} \Delta CPU_{t-i} + \varepsilon_{2t}$$
(7)

The ECM is further shown in Eq. (8) as follows:

$$\Delta REC_{t} = \theta_{1} + \sum_{i=1}^{p} \alpha_{2i} \Delta REC_{t-i} + \sum_{i=0}^{m} \alpha_{3i} \Delta \text{LNGDPC}_{t-i} + \sum_{i=0}^{n} \alpha_{4i} \Delta WTI_{t-i}$$
$$+ \sum_{i=0}^{k} \alpha_{5i} \Delta CPU_{t-i} + \gamma_{1} ECM_{t} + \varepsilon_{2t}$$
(8)

To evaluate the robustness of our estimated results, we have further employed the Fully-Modified Ordinary Least Squares (FMOLS) of [53] and the Dynamic Ordinary Least Squares (DOLS) of [54]. The estimated results are in line with the specified ARDL models.

# 4. Empirical results and discussion

#### 4.1. Baseline analyses and discussion

Before proceeding to some unit root tests, it is essential to check the descriptive statistics and correlation of the series. Table 1 shows the presence of normality and minimal heterogeneity in the data series.

Table 2 also shows the correlation matrix and does not show any threat of multi-collinearity due to the low correlation present in the series. The low variance inflation factor also confirms this finding. Interestingly, a negative and significant correlation between climate policy uncertainty and non-renewable energy demand is found. In contrast, climate policy uncertainty and renewable energy demand have a positive and significant relationship. This issue allows us to use the linear technique for modelling the pattern of renewable energy and nonrenewable energy demand functions for the United States economy. It further indicates the important role of climate policy uncertainty in the pattern of renewable and non-renewable energy demand.

This study also undertakes some unit root tests. We adopt two frequently used time series based unit root tests (the ADF, 1979 and the PP, 1988) to evaluate the stationary and non-stationary properties of the variables. Moreover, Zivot & Andrew's (2002) unit root test is applied to ensure the structural break of the examination. The unit root consequences are reported in Table 3.

Table 3 shows that the original series of each variable cannot deny the null hypothesis of the existence of a unit root across most of the variables (*REC, LNGDPC, WTI*), but their first order difference series deny the null hypothesis significantly at the 1% level. This evidence signifies that these are the first-order difference and can support cointegration examinations.

Table 4 shows the presence of the mixed-order of integration with structural breaks present in the data series. Moreover, other variables (*Non-REC, CPU*) are stationary at their levels. Finally, it shows that series signal the mixed order of integration, i.e. I(0) & I(1). These results justify an application of the ARDL bounds test to cointegration.

Table 5 shows the consequences of cointegration by applying Pesaran's cointegration approach. It is found that each test statistic significantly rejects the null hypothesis that no cointegration linkage

Table 1
Descriptive statistics.

	Non- REC	REC	LNGDPC	WTI	CPU
Mean	22.34	2.11	10.82	60.83	100.95
Median	22.31	2.15	10.81	58.08	87.25
Maximum	23.74	3.02	11.16	123.95	358.12
Minimum	19.04	1.23	10.48	20.40	17.84
Std. Dev.	0.90	0.54	0.18	25.38	70.94
Skewness	-0.57	0.12	-0.12	0.41	1.22
Kurtosis	3.63	1.57	2.04	2.26	4.33
Jarque-Bera (chi-square value)	6.21**	7.57**	3.54	4.40	27.91***
Observations	87				

Notes: LN is the natural log taken to enhance the smoothness of the data. \*\*\* and \*\* are 1% and 5% levels of significance. Non-REC, REC, GDP, WTI, and CPU are non-renewable energy consumption, renewable energy consumption, gross do-mestic product (economic growth), west texas intermediate (crude oil prices), and climate policy uncertainty. Except for real GDP per capita, all remaining variables possess seasonality in the data series. The seasonality present in the data series is adjusted with the help of additive and multiplicative de-compositions. The seasonality adjusted series are considered. Source: Authors' estimation.

#### Table 2

Pair-wise correlations and the VIF.

Variables	Non_REC	REC	LNGDPC	WTI	CPU
Non_REC	1.000				
REC	-0.548***	1.000			
LNGDPC	-0.414***	0.948***	1.000		
WTI	-0.126	0.267**	0.297**	1.000	
CPU	-0.353***	0.621***	0.674***	0.041	1.000
Average Variance	1.73				
Inflation Factor (VIF)					

Note: \*\*\* and \*\* represent 1% and 5% levels of significance. Source: Authors' estimation.

exists among the variables. This evidence indicates that these variables own stable equilibrium in the long run, and estimation of these variables can be done.

Table 6 reports the long-run estimation results of the variables in models 1 & 2. As revealed in Model 1 of Table 6, the outcome indicates that the impact of economic growth on non-renewable energy consumption is ineffective as its coefficient remains insignificant in the long run. This result suggests that even if people in the United States have income, they do not spend money buying non-renewable energy from the open market. This evidence is because they think about environmental protection and spend money on alternative energy compatible with nature.

The influence of the United States crude oil prices on non-renewable energy consumption are positive and insignificant. This result indicates

### Table 3

ADF (1979)	and PP	(1988)	unit root	test	results
		(1)00)			reourco

that even if crude oil prices rise, it does not effectively increase the nonrenewable energy demand of the people in the United States. This evidence could be because the efficiency involved in non-renewable energy sources than renewable energy sources does not convince many people in the United States. For example, consumers are habituated to the usage of fossil fuels. They know how to use and do the work quickly, but this is not the case when they go for renewable energy sources. Therefore, this motivates fewer people in the United States to demand more fossil fuels when crude oil prices rise.

The climate policy uncertainty adversely influences non-renewable energy consumption in the United States. This outcome indicates that the higher the climate policy uncertainty, the lesser demand for fossil fuels will be. This issue could be one of the reasons for the producers to reduce their fossil fuel demand when the government in the United States imposes taxes on pollution creating firms. The pollution tax imposition can come when firms use dirty energy extensively.

The first quarter of 2008 structural break also significantly reduced the demand for non-renewable energy in the United States. This result indicates that 2008 was the sub-prime crisis that originated from the banking and real estate sectors of the United States [44,55] and affected fossil fuel demand. This evidence is related to the issue because people in the United States have their income loss due to rising unemployment in the construction and banking industries. This result denies people to consumption of more fossil fuels.

As revealed in Model 2 of Table 6, the outcome indicates that economic growth positively affects renewable energy demand in the United States. One explanation might be that rising economic growth in the

	Levels		I <sup>st</sup> differences	I <sup>st</sup> differences		
	Intercept	Intercept & Trend	Intercept	Intercept & Trend		
ADF						
Non-REC	-8.087 (-3.530)	-10.156 (-4.071)	-17.212 (-3.531)	-17.112 (-4.073)	I(0)	
REC	-1.026 (-3.530)	-6.319 (-4.071)	-11.471 (-3.531)	-11.438 (-4.073)	I(1)	
LNGDPC	-0.375 (-3.530)	-2.688 (-4.071)	-10.467 (-3.531)	-10.403 (-4.073)	I(1)	
WTI	-2.167 (-3.530)	-2.133 (-4.071)	-7.307 (-3.531)	-7.274 (-4.073)	I(1)	
CPU	-3.607 (-3.530)	-5.026 (-4.071)	-13.534 (-3.531)	-13.456 (-4.073)	I(0)	
PP						
Non-REC	-8.087 (-3.530)	-10.187 (-4.071)	-34.177 (-3.531)	-34.023 (-4.073)	I(0	
REC	-1.026 (-3.530)	-6.078 (-4.071)	-16.234 (-3.531)	-16.414 (-4.073)	I(1)	
LNGDPC	-0.374 (-3.530)	-2.734 (-4.071)	-10.565 (-3.531)	-10.402 (-4.073)	I(1)	
WTI	-2.285 (-3.530)	-2.273 (-4.071)	-7.160 (-3.531)	-7.120 (-4.073)	I(1)	
CPU	-3.396 (-3.530)	-5.067 (-4.071)	-14.351 (-3.531)	-14.252 (-4.073)	I(0)	

Note: MacKinnon's approximate p-value is at a 1% significance level inside the parenthesis (). Source: Authors' estimation.

Table 4				
Zivot-Andrews	(2002)	unit root	test :	results.

	Levels			1st difference		
Non-REC	T statistics -11.4284***	Time break 2008Q1	Decision I(0)	T statistics -7.8292***	Time break 2008Q1	Decision I(0)
REC	-7.6872***	No Break## 2009Q1 2009Q1 ##	I(0)	-11.6718	2002Q1	I(0)
LNGDPC	-4.956*	2009Q1## 2008Q3 2003Q3##	I(1)	-12.673***	2019Q4	I(0)
WTI	-4.5495	2008Q3## 2014Q3	(1)	-7.729***	2008Q1	I(0)
СРИ	-5 9773 ***	2003Q4## 2008Q3## 2014Q3## 2019Q2	1(0)	-14 502***	202004	1(0)
	0.5770	2006Q3## 2015Q3##		1.002	20202.	

Notes: \*\*\* and \* represent 1% and 5% significance levels. Critical values: 0.01 = -5.57, 0.05 = -5.08, 0.1 = -4.82.

## Break points based on Bia-Perron Test.

Source: Authors' estimation

#### Table 5

ARDL bounds testing cointegration results.

Bound testing to cointegration			Diagnostic tests				
Estimated model	Lag length	Break year [dependent variable break year]	F- statistics	Jarque-Bera for Normality	Ramsey RESET Test	Heteroskedasticity Test: Breusch-Pagan-Godfrey	Breusch-Godfrey Serial Correlation LM Test:
Model-1	(1, 1, 1,	2008Q1	6.45***	2.82 (P-value:	F-statistic 2.62	0.27 (P-value: 0.9744)	2.02E-05 (P-value:
Non-REC = f	0, 1)			0.2437)	(P-Value: 0.0106)		0.9964)
(LNGDPC, WTI,							
CPU)							
Model-2	(1, 0, 3,	2009Q1	5.90***	2.07 (P-value:	2.75 (P-value:	1.07 (P-value: 0.3929)	1.54 (P-value: 0.2119)
REC = f(LNGDPC, WTL CPU)	0, 3)			0.3550)	0.0493)		
Significance Levels	Model-1 Cr	itical values $(n = 87, k = 4)$		Model-2 Critical v	alues $(n = 87, k = 4)$		
Narayan (2005)	Lower	Upper bound			Lower bound	Upper bound	
tabulated values	bound	• FF •• • • • • • • •				-rr	
10%	2.303	3.22			2.303	3.22	
5%	2.688	3.698			2.688	3.698	
1%	3.602	4.787			3.602	4.787	

Notes: \*\*\*, \*\* and \* represent at 1%, 5%, and 10% levels of significance, respectively. The lag length is in parenthesis (), n is the actual sample size, and k is the number of explanatory variables, including the dummy accommodated for the single structural break.

Source: Authors' estimation.

United States implies households have higher income levels and continue to demand more renewable energy from the open market for environmental protection. Moreover, producers with larger income support (i.e. subsidy) from the government increase the number and scale of factories and use environmental-friendly technology to expand production, resulting in tremendous renewable energy consumption. This result suggests that increasing the United States' economic growth adds to renewable energy consumption. This empirical evidence is consistent with the findings of [17] for G7 countries [18], for 18 emerging market economies [24], for 24 European countries [25], for six emerging economies [27], for 38 countries [26], for 64 countries [22], for the United States, Lin and Moubarak [60] for the Chinese economy [7], for 30 OECD countries [34], for selected emerging economies, and [35] for 20 OECD countries where they observe the positive impact of economic growth on renewable energy demand.

The coefficient of crude oil prices on renewable energy consumption is negative and significant at the 1% level. This finding can be interpreted as the effect of crude oil price inflation on clean energy demand in the United States. This result shows that consumers usually adopt a lifestyle based on affordable crude oil prices. Consumers decrease renewable energy consumption when crude oil prices rise. This finding indicates the right choice of consumers in the United States, even if nonrenewable energy is more expensive than renewable energy. This result is basically due to consumers' lifestyle choices in their day-to-day lives. Also, producers reduce their dependence on renewable energy when crude oil prices rise. This evidence is because they are already habituated to fossil fuel usage. This empirical evidence is consistent with the studies of [25] for China and Indonesia and [26] for 64 countries where they observe the adverse effect of oil prices on renewable energy demand. The empirical evidence from our analysis is also inconsistent with a few recent studies by Ref. [7] for 30 OECD economies and [35] for 20 OECD countries, where they find the promoting role of oil price on renewable energy demand. Our finding is inconsistent with a few studies [7,35] because their panel-based renewable energy demand modelling.

The impact of climate policy uncertainty on renewable energy consumption is positive and insignificant. This result indicates that the climate mitigating policy imposed by the United States policymakers does not become effective in increasing renewable energy consumption. This finding may be because accessing renewable energy is expensive for people. As long as producers are concerned, it can be a subsidy of the United States government that fails to incentivize producers to use renewable energy in production expansion.

Furthermore, the effect of structural break found in the initial quarter of 2009 on renewable energy consumption is positive and significant at a 1% level. This result indicates that this was the year of the

sub-prime financial crisis in the banking sector. The consumers who have taken loans from the banks following the banking crisis prefer to spend more on renewable energy consumption. This is because consumers know that the banking crisis triggered by the real estate builders will sustain, and the possible loan wave could benefit consumers not to return money to the banks.

However, short-run results are different from long-run results, as reported in Table 6. For instance, economic growth has a positive and significant impact on non-renewable energy but is insignificant in the long run. This evidence may be true in the short-run because people in the United States spend money buying non-renewable energy from the open market with their increased income level. This evidence reveals that people lack environmental awareness in the short run. Similarly, crude oil prices negatively and significantly impact non-renewable energy in the short run. Though people in the United States habituated to crude oil usage, increasing crude oil prices discouraged people from consuming less non-renewable energy. This issue may be because rising crude oil prices are evidence of an inflationary situation that can hurt people's financial savings in the United States.

Finally, the impact of crude oil prices on renewable energy is positive and significant in the short run. It shows the crucial role of crude oil prices in adopting renewable energy in the short run. This evidence could be true because people in the United States know that consuming more non-renewable energy at rising crude oil prices reduces their financial saving capacity and impedes the environmental quality by discharging higher amounts of carbon into the atmosphere. Therefore, increasing crude oil prices encourage people in the United States to prefer energy transitioning of renewable energy demand over nonrenewable energy.

The coefficients of all the error correction terms are negative and significant, supporting the validity of the error correction specification. More specifically, the negative and significant levels of the error correction model (i.e. convergence coefficient) are -0.377 for non-renewable energy use and -0.362 for renewable energy consumption. It justifies the presence of a long-run cointegrating relationship between the variables, and also, short-run disequilibrium for both Models 1 & 2 is corrected towards owning the long-run equilibrium. It further indicates that short-run disequilibrium correction is marginally higher (-0.377) for non-renewable energy consumption than the renewable energy demand, which is -0.362. It implies that the United States economy will take more than 268 (276) quarters to reach the long-run equilibrium of non-renewable energy demand (renewable energy consumption).

The R-squared values for Models 1 & 2 indicate that the factors included in the modelling energy demand equations explain more non-renewable energy use variation than renewable energy consumption.

#### Table 6

Estimated long-run an	d short-run	coefficients	based on	the ARDL	bounds	tests
Long-run analysis						

Model-1: Dependent variable (No	n-REC)		
Independent variables	Coefficient	Std. Error	t-Statistics
LNGDPC	0.706	1.269	0.556
WTI	0.006	0.006	1.057
CPU	-0.004*	0.002	-1.679
DU200801	-1.492***	0.465	-3.208
C	15.281	13.412	1.139
5	101201	101112	11105
Model-2: Dependent variable (RE	C)		
Independent variables			
LNGDPC	2.071***	0.206	10.054
WTI	-0.003***	0.0009	-3.662
CPU	0.0001	0.0003	0.299
DU <sub>2009Q1</sub>	0.495***	0.067	7.369
C	-20.384***	2.176	-9.365
Short-run analysis			
Model 1: Dependent variable (No	n PEC)		
Independent variables	II-I(LC)		
D(I NGDPC)	17 045***	2 598	6 558
D(WTI)	0.000**	2.590	2 284
D(DUMMY BREAK NON PEC)	0.435	0.380	1 116
D(DOMMI_DREAK_NON_REC)	0.433	0.059	6 421
Short run diagnostigs	-0.377	0.038	-0.421
p <sup>2</sup>	0 500	n <sup>2</sup>	0.490
R CE of recreasion	0.300	n <sub>a</sub>	0.462
SE OI regression	0.387	Mean of DV	0.012
D W statistics	0.539	Mean of DV	-0.013
D-w statistics	1.994		
Model 2 Dependent veriable (DE	0		
Independent variables	<b>(</b> )		
D(WTI)	0.0002	0.0006	0 529
D(WII)	-0.0003	0.0006	-0.538
D(WTI(-1))	0.002***	0.0007	2.955
D(WII(-2))	0.001**	0.0007	1.937
D(DUMMI_BREAK_REC)	0.16/	0.070	2.350
D(DUMMY_BREAK_REC(-1))	0.075	0.071	1.053
D(DUMMY_BREAK_REC(-2))	-0.148***	0.060	-2.470
ecm(-1)	-0.362***	0.058	-6.157
Short-run diagnostics		-2	
R~	0.361	Ra	0.311
SE of regression	0.055		
Std. Deviation of DV	0.066	Mean of DV	0.017
D-W statistics	1.919		

Notes: DV means the dependent variable.  $R_a^2$  indicates adjusted  $R^2$ . D-W statistics shows the Durbin–Watson statistic; it talks about the presence of autocorrelation. The value of the Durbin–Watson statistic lies between 0 and 4. If the value is 2, then there is no autocorrelation. If the value is between 0- and <2, there is a positive autocorrelation. If the value is between >2-and 4, there is a negative autocorrelation. \*\*\*, \*\*, and \* represent 1%, 5%, and 10% levels of significance. Source: Authors' estimations.

The climate policy uncertainty could be the main reason for this difference. The threat of autocorrelation in Durbin-Watson (D-W) statistics is minimal across Models 1 & 2. We also use Cumulative Sum Square (CUSUM) and CUSUM of squares (CUSUMsq) of recursive residuals proposed by Ref. [56] to check the ARDL model stability. Figs. 4 and 5 indicate the model stability at a 5% significance level for the non-renewable and renewable energy consumption. This result enables us to confirm the findings of the ARDL models shown in Table 6.

Finally, the FMOLS and the DOLS methods also check the robustness of the results. The results in Table 7 indicate similar results as noticed in Table 6. This evidence suggests that these are efficient findings that can be used by policymakers and governments of the United States and other advanced economies while mitigating climate change and global warming.

#### 4.2. Robustness checks

Robustness checks are implemented in this section by taking three steps. First, we examine whether baseline findings hold for alternative estimators, i.e. structural break unit root test and cointegration with regime shifts. Second, we test that the main results are robust to the ARDL model. We have added new structural breaks arising in the data series. Third, we also use both FMOLS and DOLS for estimation in the presence of new structural breaks.

Tables 8 and 9 present econometric methodological comparisons of structural break unit root and cointegration with regime changes. The results reported in Tables 8 and 9 confirm the robustness of our baseline estimation results. However, it is worth noting that using new structural breaks in the ARDL bounds test, short-run and long-run, as well as in FMOLS and DOLS models, does not materially alter the economic magnitude and statistical significance of coefficients on renewable energy and non-renewable energy consumption in the United States (see Tables A1, A2, and A3 of the Appendix). The evidence reported in Tables A2 and A3 confirms that climate policy uncertainty is essential to the USA economy's renewable and non-renewable energy demand.

# 5. Policy implications

The findings of this study bear important policy implications. The United States is one of the mature economies where economic growth is not the driving factor of non-renewable energy use. It shows that policymakers in the United States need to understand whether the energy impact of income level is shifting to renewable energy sources. If it happens to be true, it is a good sign for the natural environment in the United States. Crude oil prices also increase non-renewable energy demand in the long run. It shows that rising crude oil prices encourage people to use non-renewable energy. The positive impact of rising crude oil prices on non-renewable energy use is beneficial for the people in the United States because of oil efficiency and subsidy. For instance, people can buy oil because it is available in the open energy market whenever needed. Even if the oil price is much higher for the people, but still the demand for oil increases because they know that oil price could sometimes be reduced later because of the oil subsidy provided by the government to the oil producers to avoid the adverse effect on the economy [25]. Hence, these reasons motivate people to use more fuel in economic activities and limit its availability for future generations. It is also a matter of concern as excessive usage of fossil fuels can put the quality of



Fig. 4. Cumulative sum and cumulative sum of the squares at the 5% level of significance.



Fig. 5. Cumulative sum and cumulative sum of the squares at the 5% level of significance.

Table 7	
DOLS and FMOLS robust checking results.	

	NON-REC: Dependent Variable		REC: Dependent Variable	
Variable LNGDPC WTI CPU DUMMY_BREAK R <sup>2</sup>	DOLS 2.177*** [71.616] 0.006 [1.313] -0.005*** [-2.981] -1.756*** [-5.978] 0.750	FMOLS 2.178*** [94.793] 0.007** [2.004] -0.004*** [-3.480] -1.920*** [-8.316] 0.588	DOLS 0.091*** [7.870] -0.001 [-0.984] 0.004*** [6.915] 0.736*** [8.898] 0.968	FMOLS 0.130*** [12.594] -0.0009 [-0.575] 0.002*** [4.713] 0.770*** [8.466] 0.826
Adjusted R <sup>2</sup>	0.695	0.573	0.948	0.819

Notes: \*\*\* and \*\* indicate 1% and 5% levels of significance. [] shows the t-statistics. 2008Q1 is a break year for NON\_REC, and 2009Q1 is for REC. The dummy break is used for the structural break year. Lag 1 (3), decided by the Akaike Information criteria, is chosen for NON-REC (REC) variable. These lags apply only to the DOLS technique.

Source: Authors' estimations.

the natural environment at risk through increasing greenhouse gases in the atmosphere. Therefore, policymakers in the United States need to understand the role of crude oil prices in energy conservation building and mitigating climate strategies.

Interestingly, climate policy uncertainty significantly reduces nonrenewable energy use in the long run. The government imposes a higher pollution tax on producers because they use dirty energy extensively in production activity. It further indicates that increasing the pollution tax reduces the usage of non-renewable energy in the United States, which is a good signal, keeping its beneficial environmental quality in mind. Hence, the policymakers should consider the role of climate policy uncertainty in energy conservation and climate change mitigating policies.

We also find economic growth's crucial positive and significant role on renewable energy demand in the United States. This finding confirms [22] study conducted for the United States. It indicates that economic growth has become remarkable for the people as it increases their income through employment generation. As a result, people with rising income levels buy renewable energy for consumption and production activities. Similarly, the fiscal government should invest more in renewable energy capacity generation that can be supplied to the open market on a bigger scale. Thus, increased renewable energy supply over demand can reduce the price in the open energy market. Reducing renewable energy prices can motivate people in the United States to

#### Table 8

[34] Structural break unit root test.

	Levels			1st difference		
	T-statistics	Time break	Decision	T-statistics	Time break	Decision
Non-REC	-5.75***	2008Q2	I(0)	-6.41	2017Q3	I(0)
REC	-3.59	2009Q4	I(1)	-8.21	2016Q4	I(0)
LNGDPC	-4.482**	2008Q2	I(1)	-12.67***	2019Q4	I(0)
WTI	-5.19***	2014Q2	(0)	-7.67***	2014Q1	I(0)
CPU	-4.33**	2016Q2	I(0)	-7.43***	2016Q2	I(0)

Note: \*\*\*represents significance at the 1% level.

#### Table 9

Gregory and Hansen test [26] for cointegration with regime shifts.

Statistics	Variables	Test Statistic	Breakpoint	Critical Values			Cointegrated
				1%	5%	10%	
ADF	Non-REC REC	-6.07** -5.97*	2008Q3 2008Q4	-6.51	-6.00	-5.75	Yes Yes
Zt	Non-REC REC	-6.43** -6.00*	2008Q3 2008Q4	-6.51	-6.00	-5.75	Yes Yes

Note: \*\* and \* represent significance at the 5% and 10% levels, respectively.

extensively use renewable energy in both consumption and production activities. Eventually, the massive usage of renewable energy also promotes the quality of the natural environment and habituates the future generation for its use. Thus, the policymakers need to consider the role of economic growth while designing clean energy building policy for sustainable development and the environment in the long run [7].

We further confirm crude oil prices' negative and significant impact on renewable energy. It indicates that higher crude oil prices reduce renewable energy use in the United States. Even if oil prices rise, people in the United States still reduce renewable energy demand and increase their dependence on oil. The possible reason might be that people are habituated to oil usage because of its higher efficiency than renewable energy. Another reason might be that renewable energy has a limited supply and is becoming expensive for people to afford. Furthermore, oil is imported by the United States to mitigate the domestic market requirements. Therefore, the rising oil price is also subsidized by the government of the United States to avoid its adverse impact on the economy [25]. These are why people in the United States reduce their demand for renewable energy and increase the demand for non-renewable energy, which is also evident in our analysis. Therefore, the policymakers should design a policy toward making renewable energy at an affordable rate. As a result, the adoption of renewable energy should be higher by the people in the United States for a sustainable environment.

Finally, the study confirms climate price uncertainty's negative and insignificant impact on renewable energy consumption. It shows that climate policy uncertainty is ineffective in increasing renewable energy in the United States. Thus, the policymakers should provide effective financial support to enlarge the extensive demand for renewable energy because of its expensive nature. Since renewable energy is costlier than fossil fuels and beneficial for the natural environment, imposing less pollution tax on green producers should be implemented. As a result, such producers would think of producing and supplying more renewable energy into the open market. In such a way, the scale of energy production would occur massively, benefiting the producers in maximizing their profit and reducing the clean energy market uncertainty. Reducing clean energy market uncertainty reduces clean energy insecurity and poverty and promotes the quality of the natural environment [43].

## 6. Conclusion

The determinants of energy consumption patterns have been an important issue in achieving sustainable environmental quality in the current context of energy depletion, climate change, and global warming. The following facts motivated us to explore the topic. The first fact is that the concept of "green recovery" or "green economic growth" in 2020 has reignited the debate among scholars about the influence of climate policy uncertainty, economic growth and crude oil prices on the pattern of energy consumption (non-renewable and renewable) pressure. The second fact is that the relevant empirical findings are inconclusive when they are based on the influences of economic growth and crude oil prices on the energy consumption pattern. This evidence suggests that we should use climate policy uncertainty as to the key factor in non-renewable and renewable energy consumption functions, which need to be further investigated empirically using the right methods. The third fact is that advanced economies like the United States are the leading energy consumers and second-best pollution creating countries globally; experiencing the factual findings can provide effective policy recommendations for their authorities and departments to adopt energy sustainability measures. This result can promote sustainable environmental quality for the United States and other growing economies in the long run.

In this connection, we argue that this is the first study that motivates us to examine the effects of climate policy uncertainty on renewable and non-renewable energy consumption in the United States over the quarterly data from 2000Q1 to 2021Q3. As control variables, economic growth and crude oil prices are added to non-renewable and renewable energy functions. This objective is modelled by applying the ARDL bounds test to cointegration. This model helps us to estimate the longrun and short-run results. Moreover, the DOLS and the FMOLS techniques are used as results of robust checking purposes.

The empirical results from the ARDL model display the positive and insignificant impact of economic growth on non-renewable energy consumption in the United States. This finding shows that economic growth is ineffective in stimulating non-renewable energy use. Nonrenewable energy use is positively (negatively) related to crude oil prices (climate policy uncertainty). These results indicate that though rising crude oil prices enhance non-renewable energy demand but remain insignificant, the greater climate policy uncertainty reduces it. The results also show the positive and significant impact of economic growth on renewable energy use in the long run. However, rising crude oil prices reduce renewable energy. Surprisingly, climate policy uncertainty positively affects renewable energy use but is insignificant, indicating that any climate change mitigating policy does not encourage people to use clean energy in the United States.

Our study is not free from limitations. One limitation is that our theoretical framework on the impact of climate policy uncertainty on renewable and non-renewable energy consumption needs to be expanded further. Second, this study only considers the role of climate policy uncertainty on renewable and non-renewable energy consumption in the United States but neglects the effects of economic policy uncertainty and energy market uncertainty [37,43]. Interestingly, existing studies have shown that environmental regulation and technological innovation can influence the energy demand in developed and emerging economies [57-59]; Hao et al., 2018; Liu et al., 2018). Given this, future research can introduce these variables into the framework. Finally, the long-term nature of climate policy changes posed by the threat of climate change and global warming inevitably requires more research from advanced methodological viewpoints over an extended period. What is clear is that the issue of energy demand determinants that we address in this paper cannot be resolved based on aggregate empirical analysis alone [58]. Therefore, more research is needed on non-renewable and renewable energy consumption determinants in developed and emerging economies within a panel framework. As a result, the findings emerging from panel studies can have generalization possibility, which can help policymakers to design the right clean energy transition policy for mitigating climate change and global warming.

#### Authors' contributions

**Yunfeng Shang:** Writing – Original Draft; Conceptualisation, **Ding Han:** Writing – Original Draft; Final Input, **Giray Gozgor:** Writing – Original Draft; Supervision, **Mantu Kumar Mahalik:** Writing – Revised Draft; Supervision, **Bimal Kishore Sahoo:** Data Curation; Investigation; Writing – Formal Analysis, We all confirm no known conflicts of interest associated with this research, and there has been no significant financial support for this work that could have influenced its outcome.

### Declaration of competing interest

The authors declare that they have no known competing financial interests or personal relationships that could have appeared to influence the work reported in this paper.

# Table A1

ARDL Bounds Testing Cointegration Results

Bound testing to cointegration				Diagnostic tests				
Estimated models	Lag length	Break year [dependent variable break year]	F- statistics	Jarque-Bera for Normality	Ramsey RESET Test	Heteroskedasticity Test: Breusch-Pagan-Godfrey	Breusch-Godfrey Serial Correlation LM Test:	
Model-1 Non-REC = f (LNGDP, WTI, CPU)	(1, 1, 1, 0, 0)	2008Q2	6.394***	3.205 (0.201)	F-statistics: 7.436 (P-value: 0.007)	F-statistics: 0.212 (P-value: 0.98)	F-statistics: 0.099 (P- Value: 0.75)	
Model-2 Non-REC = f (LNGDP, WTI, CPU)	(1, 1, 1, 0, 0)	2008Q3	6.804***	2.727 (0.255)	F-statistics: 8.70 (P-value: 0.004)	F-statistics: 0.104 (P-value: 0.997)	F-statistics: 0.823 (P- value: 0.366)	
Model-3 REC = f(LNGDP, WTI, CPU)	(1, 0, 1, 0, 0)	2009Q4	4.40**	8.82 (0.01)	0.42 (0.65)	0.80 (0.57)	1.179 (0.31)	
Model-4 REC = f(LNGD, WTI, CPU)	(1, 0, 3, 0, 4)	2008Q4	6.11***	1.79 (0.40)	2.87 (0.02)	1.15 (0.33)	1.34 (0.26)	
Significance levels	Model1: Critical values $[n = 87, k = 4]$		Model2: Critical values $[n = 87, k = 4]$		Model3(4): Critical values $[n = 87, k = 4]$			
Narayan (2005) tabulated values	Lower bound	Upper bound	Lower bound	Upper bound	Lower bound	Upper bound		
10%	2.303	3.22	2.303	3.22	2.303(2.303)	3.22(3.22)		
5%	2.688	3.698	2.688	3.698	2.688(2.688)	3.698(3.698)		
1%	3.602	4.787	3.602	4.787	3.602(3.602)	4.787(4.787)		

Notes: P-values are in parenthesis (). The lag length is selected based on the Akaike Information Criteria. \*\*\*, \*\* and \* indicate 1%, 5%, and 10% levels, respectively. n is the actual sample size, and k is the number of explanatory variables.

# Table A2

Estimated Long-run and Short-run Coefficients Using the ARDL Bounds Test Approach

The second secon			
Long-run anarysis Model 1: Dependent veriable (Nep BEC)			
Independent variables	Coofficient	Ctub Terror	t Statistica
Independent variables	Coefficient	Sta. Error	
LNGDPC	0.900	1.229	0.732
WII	0.008	0.005	1.448
CPU	-0.003*	0.002	-1.665
DU <sub>2008Q2</sub>	$-1.639^{***}$	0.428	-3.821
C	13.195	12.982	1.016
Model-2: Dependent variable (Non-REC)			
Independent variables	Coefficient	Std. Error	t-Statistics
LNGDPC	1.133	1.149	0.986
WTI	0.006	0.004	1.279
CPU	-0.004**	0.002	-1.933
DU <sub>2008Q3</sub>	-1.670***	0.382	-4.364
С	10.822	12.123	0.892
Model-3: Dependent variable (REC)			
Independent variables	Coefficient	Std. Error	t-Statistics
LNGDPC	2.011***	0.283	7.084
WTI	-0.001*	0.001	-1.733
CPU	0.0003	0.0004	0.812
DU <sub>200904</sub>	0.435***	0.085	5.078
C	-19.774***	2.995	-6.600
Model-4: Dependent variable (REC)			
Independent variables	Coefficient	Std. Error	t-Statistics
LNGDPC	2.099	0.195	10.743
WTI	-0.003	0.0008	-3.742
CPU	9.44E-05	0.0003	0.289
DU200804	0.489	0.065	7.457
C.	-20.694	2.066	-10.014
Short-run analysis			
Model-1: Dependent variable (Non-REC)			
Independent variables	Coefficient	Std Error	t-Statistics
D(LNGDPC)	16 552	2 595	6.377
D(DFS WTI)	-0.009	0.004	-2 115
ecm(-1)	-0.385	0.060	-6 389
Short run diagnostics	-0.505	0.000	-0.009
R_coupred	0.494	Adjusted $\mathbb{R}^2$	0.482
SE of regression	0.327	Aujusicu K	0.402
Std Deviation of DV	0.530	Mean of DV	0.012
D W statistics	0.339	wean of Dy	-0.013
D-W STATISTICS	1.944		

(continued on next page)

#### Table A2 (continued)

CUSUM at 5% level	Stable		
CUSUM sq at 5% level	Stable		
Model-2: Dependent variable (Non-REC)			
Independent variables	Coefficient	Std. Error	t-Statistics
D(LNGDPC)	16.717***	2.568	6.508
D(DES_WTI)	$-0.011^{***}$	0.004	-2.677
ecm(-1)	-0.414***	0.062	-6.591
Short-run diagnostics			
R-squared	0.504	Adjusted R <sup>2</sup>	0.492
SE of regression	0.383		
Std. Deviation of DV	0.539	Mean of DV	-0.013
D-W statistics	1.856		
CUSUM at 5% level	Stable		
CUSUM SQ at 5% level	Stable		
Model-3: Dependent variable (REC)			
Independent variables	Coefficient	Std. Error	t-Statistics
D(WTI)	0.0004	0.0006	0.809
ecm(-1)	-0.270***	0.051	-5.300
Short-run diagnostics			
R <sup>2</sup>	0.22	Adjusted R <sup>2</sup>	0.21
SE of regression	0.058		
Std. Deviation of DV	0.065	Mean of DV	0.017
D-W statistics	1.940		
CUSUM at 5% level	Stable		
CUSUM sq at 5% level	Stable		
Model-4: Dependent variable (REC)			
Independent variables	Coefficient	Std. Error	t-Statistics
D(WTI)	-0.0002	0.0007	-0.319
D(WTI(-1))	0.002	0.0007	2.953
D(WTI(-2))	0.001	0.0008	1.831
D(DUMMY_Q4_2008_REC)	0.023	0.075	0.314
D(DUMMY_Q4_2008_REC(-1))	-0.018	0.074	-0.246
D(DUMMY_Q4_2008_REC(-2))	0.066	0.075	0.880
D(DUMMY_Q4_2008_REC(-3))	-0.153	0.061	-2.517
ecm(-1)	-0.389	0.062	-6.268
Short-run diagnostics			
R <sup>2</sup>	0.37	Adjusted R <sup>2</sup>	0.31
SE of regression	0.055		
Std. Deviation of DV	0.066	Mean of DV	0.018143
D-W statistics	1.726		

Notes: DV means the dependent variable.  $R_a^2$  indicates adjusted  $R^2$ . D-W statistics shows the Durbin–Watson statistic; it talks about the presence of autocorrelation. The value of the Durbin–Watson statistic lies between 0 and 4. If the value is 2, then there is no autocorrelation. There is a positive autocorrelation if the value is between 0 and 4. If the value is 2, then there is no autocorrelation. There is a positive autocorrelation if the value is between 2 and 4. \*\*\*, \*\*, and \* represent 1%, 5%, and 10% levels of significance. Source: Authors' estimation.

#### Table A3

DOLS and FMOLS Results

Variable	Model1: NON-REC: Dependent		Model2:NON-REC: Dependent		Model3:REC: Dependent Variable		Model4:REC: Dependent Variable	
	Variable [2008Q2 as break year]		Variable [2008Q3 as break year]		[2009Q4 as break year]		[2008Q4 as break year]	
LNGDPC	DOLS	FMOLS	DOLS	FMOLS	DOLS	FMOLS	DOLS	FMOLS
	2.177***	2.179***	2.181***	2.185***	0.118***	0.133***	0.081***	0.130***
WTI	0.006 [1.353]	0.007** [1.992]	0.005 [1.183]	0.005* [1.684]	-0.0009 [-0.812]	-0.001 [-0.765]	-0.0002 [-0.1763]	-0.001 [-0.632]
CPU	-0.005***	-0.004***	-0.006***	-0.005***	0.003***	0.002***	0.005***	0.002***
	[-3.188]	[-3.602]	[-3.503]	[-4.060]	[6.872]	[5.625]	[6.505]	[4.303]
DUMMY_BREAK	-1.749***	-1.922***	-1.685***	-1.854***	0.761***	0.805***	0.633***	0.769***
	[-6.435]	[-8.896]	[-6.536]	[-9.322]	[11.949]	[10.799]	[6.1703]	[7.773]
к Adjusted R <sup>2</sup>	0.76	0.63	0.77	0.65	0.96	0.87	0.98	0.80

Notes: \*\*\*, \*\* and \* indicate 1%, 5% and 10% levels of significance. [] shows the t-statistics. Lag 1 is for Models 1 & 2 and Lags 2 & 4 for Models 3 & 4, which the Akaike Information Criteria decide. These lags apply only to the DOLS technique.

# Acknowledgements

We are very indebted to the Editor-in-Chief and Guest Editors of the Journal for their valuable suggestions on the earlier draft of this paper. We are grateful to the anonymous reviewers who provided helpful comments and suggestions to significantly improve the paper's quality. We are thankful for valuable research assistance from Gupteswar Patel, who is currently a research scholar under the supervision of the fourth author.

#### References

 T. Wheeler, J. Von Braun, Climate change impacts on global food security, Science 341 (6145) (2013) 508–513.

#### Y. Shang et al.

#### Renewable Energy 197 (2022) 654-667

- [2] L. Ye, The effect of climate news risk on uncertainties, Technol. Forecast. Soc. Change 178 (2022), 121586.
- [3] E. Dogan, Analyzing the linkage between renewable and non-renewable energy consumption and economic growth by considering structural break in time-series data, Renew. Energy 99 (2016) 1126–1136.
- [4] G. Gozgor, C.K.M. Lau, Z. Lu, Energy consumption and economic growth: new evidence from the OECD countries, Energy 153 (2018) 27–34.
- [5] A. Anwar, M. Siddique, E. Dogan, A. Sharif, The moderating role of renewable and non-renewable energy in environment-income nexus for ASEAN countries: evidence from Method of Moments Quantile Regression, Renew. Energy 164 (2021) 956–967.
- [6] G. Gozgor, A new approach to the renewable energy-growth nexus: evidence from the USA, Environ. Sci. Pollut. Control Ser. 25 (17) (2018) 16590–16600.
- [7] G. Gozgor, M.K. Mahalik, E. Demir, H. Padhan, The impact of economic globalization on renewable energy in the OECD countries, Energy Pol. 139 (2020), 111365.
- [8] N. Apergis, J.E. Payne, Renewable energy consumption and economic growth: evidence from a panel of OECD countries, Energy Pol. 38 (1) (2010) 656–660.
- [9] N. Apergis, J.E. Payne, Renewable energy consumption and growth in Eurasia, Energy Econ. 32 (6) (2010) 1392–1397.
- [10] R. Inglesi-Lotz, The impact of renewable energy consumption to economic growth: a panel data application, Energy Econ. 53 (2016) 58–63.
- [11] A.N. Ajmi, R. Inglesi-Lotz, Biomass energy consumption and economic growth nexus in OECD countries: a panel analysis, Renew. Energy 162 (2020) 1649–1654.
- [12] N. Apergis, J.E. Payne, Renewable and non-renewable energy consumption-growth nexus: evidence from a panel error correction model, Energy Econ. 34 (3) (2012) 733–738.
- [13] M. Usman, D. Balsalobre-Lorente, A. Jahanger, P. Ahmad, Pollution concern during globalization mode in financially resource-rich countries: do financial development, natural resources, and renewable energy consumption matter? Renew. Energy 183 (2022) 90–102.
- [14] M. Aydin, The effect of biomass energy consumption on economic growth in BRICS countries: a country-specific panel data analysis, Renew. Energy 138 (2019) 620–627.
- [15] R. Inglesi-Lotz, E. Dogan, The role of renewable versus non-renewable energy to the level of CO<sub>2</sub> emissions: a panel analysis of sub-saharan Africa's Big 10 electricity generators, Renew. Energy 123 (2018) 36–43.
- [16] J.E. Payne, On the dynamics of energy consumption and output in the US, Appl. Energy 86 (4) (2009) 575–577.
- [17] P. Sadorsky, Renewable energy consumption, CO<sub>2</sub> emissions and oil prices in the G7 countries, Energy Econ. 31 (3) (2009) 456–462.
- [18] P. Sadorsky, Renewable energy consumption and income in emerging economies, Energy Pol. 37 (10) (2009) 4021–4028.
- [19] British Petroleum (BP), BP Statistical Review of World Energy, BP, London, 2021. June 2021.
- [20] K. Gavriilidis, Measuring Climate Policy Uncertainty, 2021. Available at SSRN, No. 3847388.
- [21] B.S. Bernanke, Irreversibility, uncertainty, and cyclical investment, Q. J. Econ. 98 (1) (1983) 85–106.
- [22] A.M. Ohler, Factors affecting the rise of renewable energy in the US: concern over environmental quality or rising unemployment? Energy J. 36 (2) (2015) 97–115.
- [23] H. Chen, Y. Shi, M. Xu, X. Zhao, Investment in renewable energy resources, sustainable financial inclusion and energy efficiency: a case of us economy, Resour. Pol. 77 (2022), 102680.
- [24] A.C. Marques, J.A. Fuinhas, J.P. Manso, Motivations driving renewable energy in European countries: a panel data approach, Energy Pol. 38 (11) (2010) 6877–6885.
- [25] R.A. Salim, S. Rafiq, Why do some emerging economies proactively accelerate the adoption of renewable energy? Energy Econ. 34 (4) (2012) 1051–1057.
- [26] A. Omri, D.K. Nguyen, On the determinants of renewable energy consumption: international evidence, Energy 72 (2014) 554–560.
- [27] M. Aguirre, G. Ibikunle, Determinants of renewable energy growth: a global sample analysis, Energy Pol. 69 (2014) 374–384.
- [28] S.R. Paramati, N. Apergis, M. Ummalla, Financing clean energy projects through domestic and foreign capital: the role of political cooperation among the EU, the G20 and OECD countries, Energy Econ. 61 (2017) 62–71.
- [29] Y. Chen, Factors influencing renewable energy consumption in China: an empirical analysis based on provincial panel data, J. Clean. Prod. 174 (2018) 605–615.
- [30] J. Jacqmin, The role of market-oriented institutions in the deployment of renewable energies: evidences from Europe, Appl. Econ. 50 (2) (2018) 202–215.
- [31] K. Schumacher, Z. Yang, The determinants of wind energy growth in the United States: drivers and barriers to state-level development, Renew. Sustain. Energy Rev. 97 (2018) 1–13.

- [32] C. Chen, M. Pinar, T. Stengos, Determinants of renewable energy consumption: importance of democratic institutions, Renew. Energy 179 (2021) 75–83.
- [33] J. Fang, G. Gozgor, M.K. Mahalik, H. Padhan, R. Xu, The impact of economic complexity on energy demand in OECD countries, Environ. Sci. Pollut. Control Ser. 28 (26) (2021) 33771–33780.
- [34] Z. Lu, M.K. Mahalik, H. Padhan, M. Gupta, G. Gozgor, Effects of age dependency and urbanization on energy demand in BRICS: evidence from the machine learning estimator, Front. Energy Res. 9 (2021), 749065.
- [35] R. Li, H. Lee, The role of energy prices and economic growth in renewable energy capacity expansion–evidence from OECD Europe, Renew. Energy 189 (2022) 435–443.
- [36] S. Mukhtarov, J.I. Mikayilov, S. Maharramov, J. Aliyev, E. Suleymanov, Higher oil prices, are they good or bad for renewable energy consumption: the case of Iran? Renew. Energy 186 (2022) 411–419.
- [37] M. Shafiullah, M.D. Miah, M.S. Alam, M. Atif, Does economic policy uncertainty affect renewable energy consumption? Renew. Energy 179 (2021) 1500–1521.
- [38] D. Borozan, Asymmetric effects of policy uncertainty on renewable energy consumption in G7 countries, Renew. Energy 189 (2022) 412–420.
   [39] Y.O. Erzurumlu, G. Gozgor, Effects of economic policy uncertainty on energy
- demail: evidence from 72 countries, J. Chin. Econ. Bus. Stud. 20 (1) (2022) 23–38.
- [40] E. Bouri, N. Iqbal, T. Klein, Climate policy uncertainty and the price dynamics of green and Brown energy stocks, Finance Res. Lett. 47 (Part B) (2022), 102740.
- [41] Q. Zeng, F. Ma, X. Lu, W. Xu, Policy uncertainty and carbon neutrality: evidence from China, Finance Res. Lett. 47 (Part B) (2022), 102771.
- [42] K. Hoang, How does corporate R&D investment respond to climate policy uncertainty? Evidence from heavy Emitter Firms in the United States. Corporate Social Responsibility and Environmental Management, Forthcoming 29 (4) (2022) 936–949, https://doi.org/10.1002/csr.2246.
- [43] B. Xu, R. Fu, C.K.M. Lau, Energy market uncertainty and the impact on the crude oil prices, J. Environ. Manag. 298 (2021), 113403.
- [44] M.K. Mahalik, H. Mallick, What causes asset price bubble in an emerging economy? Some empirical evidence in the housing sector of India, Int. Econ. J. 25 (2) (2011) 215–237.
- [45] St Louis Fed, Federal Reserve Economic Data (FRED) Database, Federal Reserve Bank of St. Louis, St. Louis, MO, 2022.
- [46] G. Gozgor, Are shocks to renewable energy consumption permanent or transitory? An empirical investigation for Brazil, China, and India, Renew. Sustain. Energy Rev. 66 (2016) 913–919.
- [47] D.A. Dickey, W.A. Fuller, Distribution of the estimators for autoregressive time series with a unit root, J. Am. Stat. Assoc. 74 (366) (1979) 427–431.
- [48] P.C. Phillips, P. Perron, Testing for a unit root in time series regression, Biometrika 75 (2) (1988) 335–346.
- [49] E. Zivot, D.W.K. Andrews, Further evidence on the great crash, the oil-price shock, and the unit-root hypothesis, J. Bus. Econ. Stat. 20 (1) (2002) 25–44.
- [50] H.H. Pesaran, Y. Shin, Generalized impulse response analysis in linear multivariate models, Econ. Lett. 58 (1) (1998) 17–29.
- [51] M.H. Pesaran, Y. Shin, R.J. Smith, Bounds testing approaches to the analysis of level relationships, J. Appl. Econom. 16 (3) (2001) 289–326.
   [52] S. Johansen, K. Juselius, Maximum likelihood estimation and inference on
- [52] S. Johansen, K. Juselius, Maximum likelihood estimation and inference on cointegration—with applications to the demand for money, Oxf. Bull. Econ. Stat. 52 (2) (1990) 169–210.
- [53] J.H. Stock, M.W. Watson, A simple estimator of cointegrating vectors in higher order integrated systems, Econometrica 61 (4) (1993) 783–820.
- [54] P.C. Phillips, B.E. Hansen, Statistical inference in instrumental variables regression with I (1) processes, Rev. Econ. Stud. 57 (1) (1990) 99–125.
- [55] H. Mallick, M.K. Mahalik, Constructing the economy: the role of construction sector in India's growth, J. R. Estate Finance Econ. 40 (3) (2010) 368–384.
- [56] R.L. Brown, J. Durbin, J.M. Evans, Techniques for testing the constancy of regression relationships over time, J. Roy. Stat. Soc. B 37 (2) (1975) 149–163.
- [57] D. Acemoglu, P. Aghion, L. Bursztyn, D. Hemous, The environment and directed technical change, Am. Econ. Rev. 102 (1) (2012) 131–166.
- [58] A.B. Jaffe, R.G. Newell, R.N. Stavins, Environmental policy and technological change, Environ. Resour. Econ. 22 (1) (2002) 41–70.
- [59] A.B. Jaffe, R.G. Newell, R.N. Stavins, Technology policy for energy and the environment, Innovat. Pol. Econ. 4 (2004) 35–68.
- [60] B. Lin, M. Moubarak, Renewable energy consumption economic growth nexus for China, Renew. Sustain. Energy Rev. 40 (2014) 111–117.
- [61] US Energy Information Administration (EIA), Renewable Energy Explained, US EIA, Washington, DC, 2021.
- [62] US Energy Information Administration (EIA), Primary Energy Consumption by Source, US EIA, Washington, DC, 2022.